Author’s response to reviews

Title: Barriers and facilitators with medication use during the transition from hospital to home: a qualitative study among patients

Authors:
Sara Daliri (s.daliri@hva.nl)
Charlotte Bekker (c.bekker@maartenskliniek.nl)
Bianca Buurman (b.m.vanes@amc.uva.nl)
Wilma Scholte op Reimer (w.j.m.scholte.op.reimer@hva.nl)
Bart van den Bemt (b.vandenbemt@maartenskliniek.nl)
Fatma Karapinar-Çarkit (f.karapinar@olvg.nl)

Version: 2 Date: 14 Mar 2019

Author’s response to reviews:

We would like to thank the editor. We respond on each comment and highlight which adjustments were made.

Kind regards, on behalf of all co-authors,

Sara Daliri

Editor Comments:

1. Ethical Approval: In the “Ethics approval and consent to participate” section of the Declarations please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.
Authors: Changed. We have included the names of the ethics committees and reference numbers in the “Ethics approval and consent to participate” section of the Declarations (lines: 493-495) and “Study design and setting” section of the Methods (lines: 120-123). “The study was approved by the local ethics committee “Adviescommissie Wetenschappelijk Onderzoek-Medisch-Ethische Commissie” (ACWO-MEC) OLVG hospital (ID WO: 15.067) and the Board of Directors BovenIJ hospital (ID WO: 5EMeh545).”

2. Authors Contributions: Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Authors: Changed. We confirm that all authors have read and approved the final manuscript. We have included this statement in the Authors’ contributions section (line: 520).

3. Figure Legend List: Please provide figure legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Authors: Changed. We have provided the heading ‘Figure Legends’ after the References (line: 664) and moved the Figure titles from the main text of the manuscript to the Figure Legends section (lines: 665-668).

4. Additional File 2, Identifying Information: We notice that you have included age, gender and hospital of the participants in Table 2. This information may compromise anonymity and requires specific consent to publish. Alternatively, please remove age from the table, or provide the ages as age ranges (e.g 70-75, 75-80) in order to ensure the participants remain anonymous.

Authors: Changed. We have removed age from Table 2 (Additional File 2) to ensure anonymity of our participants.

5. Clean Manuscript: Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please also ensure that all figures, tables, and supplementary/additional files are cited in the text.

Authors: Changed. We have uploaded our manuscript, figures, additional files as single, final and clean versions. All figures, tables, and additional files are cited in the text.